Author’s response to reviews

Title: Astragali radix Antiasthmatic decoction prevents allergic airway hyperresponsiveness and remodeling in mice

Authors:

Su Xu (xshzxs@163.com)
Bao-Ping Tian (tianbaoping1101@163.com)
Lan-Hong Zhang (lanhong1987@126.com)
Wen Hua (hw1983@126.com)
Li-Xia Xia (xialixia027@163.com)
Zhi-Hua Chen (zhihuachen2010@163.com)
Wen Li (liwenzhjz0408@163.com)
Hua-Hao Shen (huahaoshen@163.com)

Version: 2 Date: 27 May 2013

Author’s response to reviews:

Dear editor:

We submit our manuscript entitled "Astragali radix Antiasthmatic decoction prevents allergic airway hyperresponsiveness and remodeling in mice" to BMC Complementary & Alternative Medicine for publication. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal. All authors have seen the manuscript and approved to submit to your journal.

Besides, we have made changes about our paper as your suggestions. Firstly, we have add the standard procedure about the preparation of Astragali radix Antiasthmatic Decoction (AAD) in the part of Methods- Preparation of AAD. In addition, each herbs in AAD which exert potential contribution to airway inflammation or asthma were discussed in the part of Discussion.

We also have no conflict of interests.

Correspondence should be addressed to Prof. Hua-Hao Shen at the following information: Department of Respiratory and Critical Care Medicine, Second Affiliated Hospital, Zhejiang University School of Medicine, 88 Jiefang Rd., Hangzhou 310009, China.Tel.:86-571-8898-1913;Fax:86-571-8778-3729; E-mail: huahaoshen@163.com

Thank you very much for your attention and consideration to our paper. Looking forward to your decision and with kind personal regards.

Sincerely yours,

Huahao Shen